Author's response to reviews

Title: Utility and feasibility of integrating pulse oximetry into the routine assessment of young infants at first-level clinics in Karachi, Pakistan: a cross-sectional study

Authors:

Connor A Emdin (cemdin@georgeinstitute.org.uk)
Fatima Mir (fatima.mir@aku.edu)
Shazia Sultana (shazia.sultana@aku.edu)
A M Kazi (momin.kazi@aku.edu)
Anita KM Zaidi (anita.zaidi@aku.edu)
Michelle C Dimitris (michelle.dimitris@sickkids.ca)
Daniel E Roth (daniel.roth@sickkids.ca)

Version: 2 Date: 28 August 2014

Author's response to reviews:

Dear Editor,

As request, a statement of ethics approval has now been added (page 7).

Sincerely,

Connor Emdin